Author's response to reviews

Title: Construction of the influenza A virus transmission tree in a college-based population: co-transmission and interactions between influenza A viruses

Authors:

Xu-Sheng Zhang (xu-sheng.zhang@phe.gov.uk)
Daniela De Angelis (dani@mrc-bsu.cam.ac.uk)

Version: 2
Date: 28 May 2015

Author's response to reviews:

1. the following was added at line 191-192

"Ethics considerations: Data that were used in the analysis of this study were extracted from a previous study [12] and hence did not require Human Resource Ethics committee approval."

2. line numbers have been inserted.